Author’s response to reviews

Title: A data review of airway management in patients with oral cavity or oropharyngeal cancer: A single-institution experiences

Authors:

Gang Zheng (gzheng@mdanderson.org)
Lei Feng (leifeng@mdanderson.org)
Carol M Lewis (cdmlewis@mdanderson.org)

Version: 3 Date: 18 May 2019

Author’s response to reviews:

Point to point response letter to Technical Comments:

1. Upon resubmission, please remove any tracked changes or highlighting and include only a single clean copy of the manuscript. Please ensure the document is in the final form for publication; please upload only files that are to be published and remove any files that should not form part of the final manuscript (e.g. covering letters, responses to reviewers, etc). If you wish to respond to these revision requests, please insert the information in to the designated input box only.

Response: All the tracked changes or highlighting in the manuscript are removed. The manuscript is in the final form for publication. The manuscript does not contain any files that are not for publication.

2. Please move the statement regarding consent from the "Consent to publish" to the "Ethics approval and consent to participate”. The "Consent to publish" section should contain a statement of "Not applicable”.

Response: The statement regarding consent from the “Consent to publish” is removed and added to the “Ethics approval and consent to participate”. “Not applicable” now added to the “Consent to publish”.

3. In the Authors Contributions section each author should also be specifically mentioned by their unique initials rather than their full names. Also, please include a statement in the Authors'
contributions section to the effect that all authors have read and approved the final manuscript, and ensure that this is the case.

Response: Each of authors’ full name is replaced by an initials of the first name followed by the last name; and the statement of “all authors have read and approved the final manuscript and ensure that this is the case.”